Specific Comments to Authors: Please check whether the references is complete. For instance, "ROS and RNS, causing DNA fragmentation, membrane disintegration and protein misfolding, activate the immune system."

Reply:

ROS and RNS, causing DNA fragmentation, membrane disintegration and protein misfolding, activate the immune system; immune cells and other stromal components produce inflammatory cytokines and chemokines which, together with ROS and RNS, cause epithelial cell damage and increased proliferation [10].